\begin{thebibliography}{BGHT99}

\bibitem[Ale23]{alexander}
J.~Alexander.
\newblock A lemma on a system of knotted curves.
\newblock {\em Proc. Nat. Acad. Sci. USA}, 9:93--95, 1923.

\bibitem[Ber16]{bergeron-open}
F.~Bergeron.
\newblock {Open Questions} for operators related to {Rectangular Catalan
  Combinatorics}.
\newblock arXiv:1603.04476, March 2016.

\bibitem[BGHT99]{bght-positivity}
F.~Bergeron, A.~M. Garsia, M.~Haiman, and G.~Tesler.
\newblock Identities and positivity conjectures for some remarkable operators
  in the theory of symmetric functions.
\newblock {\em Meths. and Appls. of Analysis}, 6(3):363--420, 1999.

\bibitem[BGLX15]{rational-shuffle}
F.~Bergeron, A.~Garsia, E.~S. Leven, and G.~Xin.
\newblock {Compositional} $(km, kn)$-{Shuffle} {Conjectures}.
\newblock {\em Int. Math. Research Notices}, October 2015.

\bibitem[CM15]{carlsson-mellit}
E.~Carlsson and A.~Mellit.
\newblock A proof of the shuffle conjecture.
\newblock arXiv:math/1508.06239, August 2015.

\bibitem[EH16]{elias-hogancamp}
B.~Elias and Matthew Hogancamp.
\newblock On the computation of torus link homology.
\newblock arXiv:1603.00407, March 2016.

\bibitem[GN15]{gorsky-negut}
E.~Gorsky and A.~Negut.
\newblock Refined knot invariants and {Hilbert} schemes.
\newblock {\em J. Math. Pures Appl.}, 9(104):403--435, 2015.

\bibitem[GORS14]{gors-knots}
E.~Gorsky, A.~Oblomkov, J.~Rasmussen, and V.~Shende.
\newblock Torus knots and the {Rational DAHA}.
\newblock {\em Duke Math. J.}, 163(14):2709--2794, 2014.

\bibitem[Hag08]{haglund-book}
J.~Haglund.
\newblock {\em The $q,t$-{Catalan} Numbers and the Space of Diagonal
  Harmonics}.
\newblock Amer. Math. Soc., 2008.
\newblock Vol. 41 of University Lecture Series.

\bibitem[Hag16]{haglund-knots}
J.~Haglund.
\newblock The combinatorics of knot invariants arising from the study of
  {Macdonald} polynomials.
\newblock In A.~Beveridge, J.~R. Griggs, L.~Hogben, G.~Musiker, and P.~Tetali,
  editors, {\em Recent Trends in Combinatorics}, pages 579 -- 600. The IMA
  Volumes in Math. and its Applications, 2016.

\bibitem[HHL05]{hhl}
J.~Haglund, M.~Haiman, and N.~Loehr.
\newblock A combinatorial formula for {Macdonald} polynomials.
\newblock {\em J. Amer. Math. Soc.}, 18:735--761, 2005.

\bibitem[Kho07]{khovanov}
Mikhail Khovanov.
\newblock Triply-graded link homology and {Hochschild} homology of {Soergel}
  bimodules.
\newblock {\em Internat. J. Math.}, 18(8):869--885, 2007.

\bibitem[KR08]{khovanov-rozansky}
Mikhail Khovanov and Lev Rozansky.
\newblock Matrix factorizations and link homology.
\newblock {\em Fund. Math.}, 199(1):1--91, 2008.

\bibitem[LLT97]{llt}
A.~Lascoux, B.~Leclerc, and J.-Y. Thibon.
\newblock Ribbon tableaux, {Hall-Littlewood} functions, quantum affine
  algebras, and unipotent varieties.
\newblock {\em J. Math. Phys.}, 38(2):1041--1068, 1997.

\bibitem[{Mel}16]{mellit-rational}
A.~{Mellit}.
\newblock {Toric braids and $(m,n)$-parking functions}.
\newblock arXiv:1604.07456, April 2016.

\bibitem[Ren15]{ren}
Q.~Ren.
\newblock Ordered partitions and drawings of rooted plane trees.
\newblock {\em Discrete Math.}, 338:1--9, 2015.

\bibitem[{Ser}16]{leven-square}
E.~{Sergel Leven}.
\newblock {A proof of the Square Paths Conjecture}.
\newblock arXiv:1601.06249, January 2016.

\bibitem[Slo]{oeis}
N.~J.~A. Sloane.
\newblock {The On-Line Encyclopedia of Integer Sequences}.
\newblock Published electronically at \texttt{http://oeis.org}.

\bibitem[Sta99]{ec2}
R.~P. Stanley.
\newblock {\em Enumerative Combinatorics}, volume~2.
\newblock Cambridge University Press, 1999.

\end{thebibliography}

\documentclass[submission]{FPSAC2017}

\articlenumber{38}
\addbibresource{38_Wilson.bib}

\usepackage{amsmath, tikz, ytableau}

\usepackage[latin1]{inputenc}
\usepackage{subfigure}
\usepackage{bm}

%my commands
\newcommand{\sym}{\operatorname{Sym}}
\newcommand{\sg}{\sigma}
\newcommand{\fig}[2]{\begin{figure}[ht]
\centerline{\scalebox{.66}{\epsfig{file=#1.eps}}}
\caption{#2}
\label{fig:#1}
\end{figure}}
\def\des{\operatorname{des}}
\def\Des{\operatorname{Des}}
\def\asc{\operatorname{asc}}
\def\Asc{\operatorname{Asc}}
\def\inv{\operatorname{inv}}
\def\Inv{\operatorname{Inv}}
\def\maj{\operatorname{maj}}
\def\minimaj{\operatorname{minimaj}}
\def\Maj{\operatorname{Maj}}
\def\Obj{\operatorname{Obj}}
\def\stat{\operatorname{stat}}
\def\coinv{\operatorname{coinv}}
\def\comaj{\operatorname{comaj}}
\def\dist{D}
\newcommand{\dists}[3] {D^{#1}_{#2}(#3)}
\newcommand{\distosp}[4] {D^{#1}_{#2, #3}(#4)}
\def\blocks{\operatorname{blk}}
\def\rlmaj{\operatorname{rlmaj}}
\def\rlcomaj{\operatorname{rlcomaj}}
\def\id{\operatorname{id}}
\newcommand{\qbinom}[2] { {#1 \brack #2}_{q} }
\newcommand{\qtbinom}[2] {{#1 \brack #2}_{q,t}}
\newcommand{\qint}[1] { [#1]_{q} }
\newcommand{\qtint}[1] { [#1]_{q,t} }
\newcommand{\rmaj}[1] {#1 \hspace{-0.02in} \operatorname{-maj}}
\newcommand{\abs}[1] {#1_{\operatorname{abs}}}
\newcommand{\stir}[2] {S_{#1, #2}}
\newcommand{\qstir}[3] {S_{#1, #2}(#3)}
\newcommand{\allqstir}[4] {S_{#1, #2}^{(#4)}(#3)}
\newcommand{\setp}[2] {\mathcal{P}_{#1, #2}}
\def\set{\operatorname{set}}
\def\setplus{\operatorname{set}^{+}}
\newcommand{\osp}[2] {\mathcal{OP}_{#1, #2}}
\newcommand{\ossp}[2] {\mathcal{OSP}_{#1, #2}}
\newcommand{\omp}[2] {\mathcal{OMP}_{#1,#2}}
\def\bl{\operatorname{bl}}
\newcommand{\ospn}[1] {\mathcal{OP}_{#1}}
\newcommand{\osspn}[1] {\mathcal{OSP}_{#1}}
\newcommand{\ospcount}[2] {\operatorname{OP}_{#1,#2}}
\newcommand{\eul}[2] {A_{#1, #2}}
\newcommand{\unc}{\operatorname{unc}}
\newcommand{\op}[1] {#1}
\def\S{\mathfrak{S}}
\def\P{\mathcal{P}}
\def\rook{\mathcal{RP}}
\def\stair{\operatorname{St}_{n}}
\def\multiset#1#2{\ensuremath{\left(\kern-.3em\left(\genfrac{}{}{0pt}{}{#1}{#2}\right)\kern-.3em\right)}}
\newcommand{\insertion}[3] {^{#1}_{#2, #3}}
\newcommand{\qmah}[3] {\dist_{#1, #2}(#3)}
\def\Falls{>^{\prime}}
\def\syt{\operatorname{SYT}}
\def\ssyt{\operatorname{SSYT}}
\def\Dinv{\operatorname{Dinv}}
\def\dinv{\operatorname{dinv}}
\def\area{\operatorname{area}}
\def\shuffle{\operatorname{Sh}}
\def\less{<_{\mathcal{A}}}
\newcommand{\wpf}{\mathcal{WPF}}
\newcommand{\pf}{\mathcal{PF}}
\newcommand{\bpf}{\mathcal{BPF}}
\newcommand{\pfcount}{\operatorname{PF}}
\newcommand{\ldyck}{\mathcal{LD}}
\newcommand{\dense}{\operatorname{Dense}}
\newcommand{\ddyck}{\mathcal{L}^{\operatorname{Dense}}}
\newcommand{\diagh}{\mathcal{DH}}
\newcommand{\diagr}{\mathcal{DR}}
\newcommand{\frob}{\operatorname{Frob}}
\def\sch{\mathcal{S}}
\def\schpoly{S}
\def\rs{RS}
\def\rscal{\mathcal{RS}}
\def\vs{VS}
\def\vscal{\mathcal{VS}}
\def\twocar{\mathcal{T}}
\def\twocarpoly{T}
\def\rt{RT}
\def\rtcal{\mathcal{RT}}
\def\vt{VT}
\def\vtcal{\mathcal{VT}}
\def\ret{\operatorname{ret}}
\def\dec{\operatorname{dec}}
\def\cox{\operatorname{Cox}}
\def\shi{\operatorname{Shi}}
\def\ish{\operatorname{Ish}}
\def\regions{\mathcal{R}}
\def\lab{\el}
\def\touch{\operatorname{touch}}
\def\ch{\operatorname{char}}

\newcommand\tes[3]{\operatorname{Tes}_{#1}(#2,#3)}
\newcommand\tesset[1]{\mathcal{T}_{#1}}
\newcommand\tessetprime[1]{\mathcal{T}^{\prime}_{#1}}
\newcommand\ptesset[1]{\mathcal{PT}_{#1}}
\newcommand{\vect}[1] {\mathbf{#1}}
\def\wt {\operatorname{wt}}
\def\pos{\operatorname{entries}^{+}}
\def\posrows{\operatorname{rows}^{+}}
\def\neg{\operatorname{entries}^{-}}
\def\negrows{\operatorname{rows}^{-}}
\def\alte{g}
\def\hthilb{F}
\def\hilb{\operatorname{Hilb}}
\def\vhilb{\widetilde{\operatorname{Hilb}}}
\def\sort{\operatorname{sort}}
\def\reverse{\operatorname{reverse}}
\def\hooks{\alpha}
\def\hooksvec{\operatorname{hooks}}
\def\nonzero{\operatorname{nonzero}}
\def\entries{\operatorname{entries}}
\def\rows{\operatorname{rows}}
\def\iDes{\operatorname{iDes}}
\def\hiDes{\operatorname{hiDes}}
\def\wiDes{\operatorname{aiDes}}
\def\qstat{\operatorname{qstat}}
\def\tstat{\operatorname{tstat}}
\def\Image{\operatorname{Im}}
\newcommand{\dyck} { \mathcal{D}}
\newcommand{\ld} {\mathcal{LD}}
\def\rise{\operatorname{Rise}}
\def\fall{\operatorname{Fall}}
\def\val{\operatorname{Val}}
\def\peak{\operatorname{Peak}}
\def\risepoly{\rise}
\def\valpoly{\val}
\def\comp{\operatorname{comp}}
\def\miniperm{\gamma}
\def\shape{\operatorname{shape}}
\newcommand{\diag} { \operatorname{Diag}}
\def\height{h}
\def\col{c}
\def\width{w}
\def\hinv{\operatorname{hdinv}}
\def\Hinv{\operatorname{hDinv}}
\def\winv{\operatorname{wdinv}}
\def\Winv{\operatorname{wDinv}}
\def\dinvi{d}
\def\areai{a}
\newcommand{\stack}{\operatorname{Stack}}
\newcommand{\blank}{\operatorname{Blank}}
\def\llt{LLT}
\def\word{w}
\def\cat{\operatorname{Cat}}
\def\catmod{\overline{\cat}}
\def\bi{b}
\def\cpoly{C}
\def\el{\ell}
\def\qsym {\operatorname{QSym}}
\def\cycle{\operatorname{cycle}}
\def\stair{\operatorname{Stair}}
\newcommand\mixed[2]{\mathcal{M}_{#1, #2}}
\def\image{\operatorname{image}}
\def\arm{\operatorname{arm}}
\def\leg{\operatorname{leg}}
\def\hagpoly{\pfcount^{\operatorname{h}}}
\def\wilspoly{\pfcount^{\operatorname{w}}}
\def\areapoly{\pfcount^{\area}}
\def\dinvpoly{\pfcount^{\dinv}}
\newcommand{\tail}{\operatorname{tail}}
\def\target{\operatorname{target}}
\def\surj{\mathcal{O}}
\def\car{\operatorname{car}}
\def\spot{\operatorname{spot}}
\def\considerate{\operatorname{cons}}
\def\consideratepf{\mathcal{CPF}}
\def\touch{\operatorname{touch}}
\def\spann{\operatorname{span}}
\def\dof{\operatorname{dof}}
\def\ceiling{\operatorname{ceiling}}
\def\border{\mathcal{B}}
\def\key{\mathcal{K}}
\def\keyposet{\mathbb{K}}
\def\atom{\mathcal{A}}
\def\atomposet{\mathbb{A}}
\def\I{I}
\def\M{M}
\def\wp{\mathcal{W}}
\def\linkpoly{L}
\def\linknorm{\widetilde{\linkpoly}}
\def\fubini{\overline{\mathcal{F}}}
\def\barstat{\operatorname{bar}}
\def\fulltwist{\operatorname{FT}}
\def\braidgroup{\operatorname{Br}}
\def\south{\operatorname{south}}
\def\hook{\operatorname{hook}}
\def\weight{\operatorname{wt}}
\def\itab{\operatorname{IFT}}
\def\armset{\operatorname{Arm}}
\def\chromatic{X}
\def\gtab{\operatorname{GT}}

\newtheorem{lemma}{Lemma}[section]
\newtheorem{prop}{Proposition}[section]
\newtheorem{cor}{Corollary}[section]
\newtheorem{thm}{Theorem}[section]
\newtheorem{conj}{Conjecture}[section]
\newtheorem{prob}{Open Problem}
\newtheorem{identity}{Identity}[section]
\newtheorem{project}{Project}[section]
\newtheorem{defn}{Definition}[section]
\newtheorem{example}{Example}[section]


\author{Andrew Timothy Wilson\thanks{\href{mailto:andwils@math.upenn.edu}{andwils@math.upenn.edu}. The author was supported by an NSF Mathematical Sciences Postdoctoral Research Fellowship.}}
\title{Torus link homology and the nabla operator}
\address{Department of Mathematics, University of Pennsylvania}
\keywords{torus links, symmetric functions, Macdonald polynomials, nabla operator}
\received{\today}
\revised{\today}
\accepted{tomorrow}


\abstract{
In recent work, Elias and Hogancamp develop a recurrence for the Poincar\'e series of the triply graded Hochschild homology of certain links, one of which is the $(n,n)$ torus link. In this case, Elias and Hogancamp give a combinatorial formula for this homology that is reminiscent of the combinatorics of the modified Macdonald polynomial eigenoperator $\nabla$. We give a combinatorial formula for the homologies of all links considered by Elias and Hogancamp. Our first formula is not easily computable, so we show how to transform it into a computable version. Finally, we conjecture a direct relationship between the $(n,n)$ torus link case of our formula and the symmetric function $\nabla p_{1^n}$.
}

\begin{document}

\maketitle

\section{Introduction}

We begin by establishing some notation from knot theory, following \cite{elias-hogancamp}. The remaining sections of the paper will take a more combinatorial perspective.

The \emph{braid group on $n$ strands}, denoted $\braidgroup_n$, can be defined by the presentation
\begin{align}
\braidgroup_n &= \left\langle \sg_1, \sg_2, \ldots, \sg_{n-1} \, | \, \sg_i \sg_{i+1} \sg_i = \sg_{i+1} \sg_i \sg_{i+1}, \, \sg_i \sg_j = \sg_j \sg_i \right\rangle
\end{align}
for all $1 \leq i \leq n-2$ and $|i-j| \geq 2$. This group can be pictured as all ways to ``braid'' together $n$ strands, where $\sg_i$ corresponds to crossing string $i+1$ over string $i$ and the group operation is concatenation. One particularly notable braid is the \emph{full twist braid} on $n$ strands, denoted $\fulltwist_n$, which can be written
\begin{align}
\fulltwist_n &=\left( (\sg_1) (\sg_2 \sg_1) \ldots (\sg_{n-1} \sg_{n-2} \ldots \sg_1) \right)^2.
\end{align}
where multiplication is left to right. We will also need an operation $\omega$ on braids which corresponds to rotation around the horizontal axis. We define $\omega$ on $\braidgroup_n$ by $\omega(\sg_i) = \sg_i$ and $\omega(\alpha \beta) = \omega(\beta) \omega(\alpha)$. Then $\omega$ is an anti-involution on $\braidgroup_n$. All of our braids will have the property that the string that begins in column $i$ also ends in column $i$ for all $i$; these are sometimes called \emph{perfect braids}.

Given a braid with $n$ strands, one can form a \emph{link} (i.e.\ nonintersecting collection of knots) by identifying the top of the strand that begins in position $i$ with the bottom of the strand that ends in position $i$ for $1 \leq i \leq n$. The result is called a \emph{closed braid}. Alexander proved that every link can be represented by a closed braid (although this representation is not unique) \cite{alexander}. The closure of a perfect braid is a link that consists of $n$ separate unknots linked together.

In \cite{elias-hogancamp}, Elias and Hogancamp assign a complex $C_v$ to every binary word $v$. We describe this assignment here -- see \cref{fig:cv-braid} for an example. Say $v \in \{0,1\}^n$ with $|v| = m$. We begin with two braids, the full twist braid $\fulltwist_{n-m}$ and a certain recursively defined complex $K_m$ \cite{elias-hogancamp}, which sits to the right of $\fulltwist_{n-m}$. For $i = 1$ to $n$, we feed  string $i$ into the leftmost available position in $K_m$ if $v_i = 1$; otherwise, we feed string $i$ into the leftmost available position in $\fulltwist_{n-m}$. All crossings that occur are forced to be ``positive,'' i.e.\ the right strand crosses over the left strand. This induces a braid $\beta_v \in \braidgroup_n$ that occurs before the adjacent $\fulltwist_{n-m}$ and $K_m$. The final complex $C_v$ is obtained by performing $\omega(\beta_v)$, followed by $\beta_v$, followed by the adjacent $\fulltwist_{n-m}$ and $K_m$. We note that $C_{0^n}$ is the full twist braid $\fulltwist_n$ and that the closure of this braid is the $(n,n)$ torus link. The combinatorics of other links, in particular the $(m,n)$ torus link for $m$ and $n$ coprime, has been studied by a variety of authors in recent years \cite{gors-knots, gorsky-negut}. Haglund gives an overview of this work from a combinatorial perspective in \cite{haglund-knots}.

\begin{figure}
\centering
\includegraphics[scale=1]{afigure}
\caption{We have drawn the complex $C_{10101101}$, where $\fulltwist_3$ is the full twist braid and $K_5$ is a certain complex defined recursively in \cite{elias-hogancamp}. This figure is used courtesy of \cite{elias-hogancamp}.}
\label{fig:cv-braid}
\end{figure}


Elias and Hogancamp map each complex $C_v$ to a graded Soergel bimodule and then consider the \emph{Hochschild homology} of this bimodule; this is sometimes called Khovanov-Rozansky homology \cite{khovanov, khovanov-rozansky}.
This homology has three gradings: the bimodule degree (using the variable $Q$), the homological degree ($T$), and the Hochschild degree ($A$). After the grading shifts $q = Q^2$, $t = T^2 Q^{-2}$, and $a = AQ^{-2}$, Elias and Hogancamp give a recurrence for the Poincar\'e series of this triply graded homology, which they denote $f_{v}(q,a,t)$. They also give a combinatorial formula for the special case $f_{0^n}(q,a,t)$. We will give two combinatorial formulas for $f_{v}(q,a,t)$ for every $v \in \{0,1\}^n$. 

In \cref{sec:infinite}, we define a symmetric function $\linkpoly_v(x;q,t)$ which we call the \emph{link symmetric function}. Its definition is reminiscent of the combinatorics of the Macdonald eigenoperator $\nabla$, introduced in \cite{bght-positivity}. We prove that $f_v(q,a,t)$ is equal to a certain inner product with $\linkpoly_v(x;q,t)$. 

The main weakness of our first formula is that it is a sum over infinitely many objects, so it is not clear how to compute using this formula. We address this issue in \cref{sec:finite}, obtaining a finite formula for $L_v(x;q,t)$ using a collection of combinatorial objects we call \emph{barred Fubini words}.

We close by presenting some conjectures in \cref{sec:conjectures}. In particular, we conjecture that
\begin{align}
\linkpoly_{0^n}(x;q,t) &= (1-q)^{-n} \nabla p_{1^n} .
\end{align}
where the terminology is defined in \cref{sec:conjectures}.
A proof of this conjecture would provide the first combinatorial interpretation for $\nabla p_{1^n}$. There has been much recent work establishing combinatorial interpretations for $\nabla e_n$ \cite{carlsson-mellit} and $\nabla p_n$ \cite{leven-square}. We believe that $\linkpoly_v(x;q,t)$ is also related to Macdonald polynomials for general $v$, although we do not have an explicit conjecture in this direction.

\section{An infinite formula}
\label{sec:infinite}

Let $\mathbb{N} = \{0,1,2,\ldots\}$ and $\mathbb{Z}_{+} = \{1,2,3,\ldots\}$. We begin by defining two statistics. 
 
\begin{defn}
Given words $\gamma \in \mathbb{N}^n$ and $\pi \in \mathbb{Z}_{+}^n$, we define
\begin{align}
\area(\gamma) &= |\gamma| - \# \{1 \leq i \leq n : \gamma_i > 0 \} \\
\dinv(\gamma, \pi) &= \# \{1 \leq i < j \leq n : \gamma_i = \gamma_j, \pi_i > \pi_j \} \\
&+ \#\{1 \leq i < j \leq n : \gamma_i + 1 = \gamma_j, \pi_i < \pi_j \} \nonumber \\
x^{\pi} &= \prod_{i=1}^{n} x_{\pi_i} .
\end{align}
\end{defn}

In \cref{fig:area-dinv-example}, we draw a diagram for $\gamma = 20141022$ and $\pi = 41322231$. Area counts the empty boxes in such a diagram, dinv counts certain pairs of labels, and $x^{\pi}$ records all labels that appear in the diagram. 

\begin{figure}
\begin{center}
\begin{tikzpicture}
\draw [step=0.5cm] (-0.001, 0) grid (0.5, 1);
\draw [step=0.5cm] (0.499, 0) grid (1, 0);
\draw [step=0.5cm] (0.999, 0) grid (1.5, 0.5);
\draw [step=0.5cm] (1.499, 0) grid (2, 2);
\draw [step=0.5cm] (1.999, 0) grid (2.5, 0.5);
\draw [step=0.5cm] (2.499, 0) grid (3, 0);
\draw [step=0.5cm] (2.999, 0) grid (3.5, 1);
\draw [step=0.5cm] (3.499, 0) grid (4, 1);

\node at (0.25, 0.75) {4};
\node at (0.75, -0.25) {1};
\node at (1.25, 0.25) {3};
\node at (1.75, 1.75) {2};
\node at (2.25, 0.25) {2};
\node at (2.75, -0.25) {2};
\node at (3.25, 0.75) {3};
\node at (3.75, 0.75) {1};

\end{tikzpicture}
\end{center}
\caption{We have depicted the example $\gamma = 20141022$ and $\pi = 41322231$ by drawing bottom-justified columns with heights $\gamma_1$, $\gamma_2$, \ldots, $\gamma_8$ and the labels $\pi_i$ are placed as high as possible in each column. In this example, we compute $\area(\gamma) = 6$, $\dinv(\gamma, \pi) = 7$, where the contributing pairs are in columns $(1, 7)$, $(1, 8)$, $(2,3)$, $(2,5)$, $(3, 5)$, $(5, 7)$, $(7, 8)$, and $x^{\pi} = x_1^2 x_2^3 x_3^2 x_4$.}
\label{fig:area-dinv-example}
\end{figure}



\begin{defn}
Given $n \in \mathbb{Z}_{+}$ and $v \in \{0,1\}^n$, define 
\begin{align}
\linkpoly_{v} &= \linkpoly_{v}(x; q, t) = \sum_{\substack{\gamma \in \mathbb{N}^n, \, \pi \in \mathbb{Z}_{+}^n \\ \gamma_i = 0 \Leftrightarrow v_i = 1}}  q^{\area(\gamma)} t^{\dinv(\gamma, \pi)}x^{\pi} .
\end{align}
\end{defn}

Perhaps the first thing to note about $\linkpoly_{v}$ is that it is symmetric in the $x$ variables; one way to see this is to express $\linkpoly_v$ as a sum of LLT polynomials \cite{llt}. More precisely, each $\gamma \in \mathbb{N}^n$ can be associated with an $n$-tuple $\boldsymbol{\lambda}(\gamma)$ of single cell partitions in the plane, where the $i$th cell is placed on diagonal $\gamma_i$ and the order is not changed. Using the notation of \cite{hhl}, the unicellular LLT polynomial $G_{\boldsymbol{\lambda}(\gamma)}(x; t)$ can be used to write
\begin{align}
\linkpoly_{v} &= \sum_{\substack{\gamma \in \mathbb{N}^n \\ \gamma_i = 0 \Leftrightarrow v_i = 1}} q^{\area(\gamma)} G_{\boldsymbol{\lambda}(\gamma)}(x; t) .
\end{align}
Since LLT polynomials are symmetric, every $L_v$ is also symmetric.

We also remark that $\linkpoly_{1^n}$ is equal to the modified Macdonald polynomial $\widetilde{H}_{1^n}(x;q,t)$, which is also equal to the graded Frobenius series of the coinvariants of $\S_n$ with grading in $t$.

Next, we note that the Poincar\'e series $f_v(q,a,t)$ can be recovered as a certain inner product of $\linkpoly_v$. We follow the standard notation for symmetric functions and their usual inner product, as described in Chapter 7 of \cite{ec2}. The following lemma leads to \cref{thm:fv-lv}. We omit the proofs from this extended abstract.

\begin{lemma}
\label{lemma:first1}
\begin{align}
\linkpoly_{0^n} &= \frac{1}{1-q} \linkpoly_{10^{n-1}} .
\end{align} 
\end{lemma}


\begin{thm}
\label{thm:fv-lv}
For any $v \in \{0,1\}^n$,
\begin{align}
f_{v}(q, a, t) &= \sum_{d=0}^{n} \left\langle \linkpoly_{v}, e_{n-d} h_d \right\rangle a^d .
\end{align}
\end{thm}



For the sake of comparison with \cite{elias-hogancamp}, we give a simplified formula that directly computes $f_v(q,a,t)$ from \cref{thm:fv-lv}. Given $\gamma \in \mathbb{N}^n$ and $1 \leq i \leq n$, let 
\begin{align}
\dinv_i(\gamma) &= \# \{j < i : \gamma_j = \gamma_i\} + \# \{j > i : \gamma_j  = \gamma_i + 1 \}.
\end{align}

\begin{cor}
\label{cor:f-infinite}
\begin{align}
f_v(q,a,t) &= \sum_{\substack{\gamma \in \mathbb{N}^n \\ \gamma_i = 0 \Leftrightarrow v_i = 1}} q^{\area(\gamma)} \prod_{i=1}^{n} \left(a + t^{\dinv_i(\gamma)} \right) 
\end{align}
where, as before, $\area(\gamma) = |\gamma| - \# \{1 \leq i \leq n : \gamma_i > 0 \}$. 
\end{cor}

If $v=0^n$ and $a=0$, this is exactly Theorem 1.9 in \cite{elias-hogancamp}. 

Let us use \cref{cor:f-infinite} to compute $f_{00}(q,a,t)$. We consider three cases: either $\gamma_2 \in \{\gamma_1, \gamma_1 +1\}$, $\gamma_2 > \gamma_1 + 1$, or $\gamma_2 < \gamma_1$. In the first case, we get a contribution of $(1+a)(a+t)$ from the dinv product in \cref{cor:f-finite}; in the other cases, we get $(1+a)^2.$ 

In the first case, our minimum area is either 0 or 1, depending on if $\gamma_2 = \gamma_1$ or $\gamma_1 + 1$. Either way, we can choose any nonnegative integer $k$ and add $k$ to both $\gamma_1$ and $\gamma_2$. This yields an area contribution of  $(1+q)(1-q^2)$. In the second case, the minimum area is 2 and not only can we add area to both $\gamma_1$ and $\gamma_2$ equally but we can add area to $\gamma_2$ by itself. The resulting contribution is $q^2/((1-q^2)(1-q))$. Similarly, in the third case we get a contribution of $q/((1-q^2)(1-q))$. 

Summing up all these weights, we get
\begin{align}
f_{00}(q,a,t) &= \frac{(1+q)(1+a)(a+t)}{1-q^2} + \frac{q^2(1+a)^2}{(1-q^2)(1-q)} + \frac{q(1+a)^2}{(1-q^2)(1-q)} \\
&= \frac{q+t-qt + a(1+q+t-qt) +a^2}{(1-q)^2} .
\end{align}
This process can become difficult for $v$ with length as short as 3. The difficulty comes mostly from considering how to add (possibly infinite amounts of) area without affecting the dinv vectors. We address this issue in the next section by deriving a formula that allows us to restrict to a finite set of $\gamma$ vectors.

\section{A finite formula}
\label{sec:finite}

Although the combinatorial definition of $\linkpoly_{v}$ is straightforward, it is not computationally effective\footnote{There are also infinitely many $\pi \in \mathbb{Z}_{+}^n$, but this problem can be addressed with standardization \cite{haglund-book}.} since it is a sum over infinitely many words $\gamma \in \mathbb{N}^n$. We rectify this issue in \cref{thm:link-finite} below. The idea is to compress the vectors $\gamma$ while altering the statistics so that the link polynomial $\linkpoly_v$ is not changed.

\begin{defn}
A word $\gamma \in \mathbb{N}^n$ is a \emph{Fubini word} if every integer $0 \leq k \leq \max(\gamma)$ appears in $\gamma$. \end{defn}

For example, $41255103$ is a Fubini word but $20141022$ is not a Fubini word, since it contains a 4 but not a 3. We call these Fubini words because they are counted by the Fubini numbers (\cite{oeis}, A000670), which also count ordered partitions of the set $\{1,2,\ldots,n\}$. We will actually be interested in certain decorated Fubini words.

\begin{defn}
Given $v \in \{0,1\}^n$, we say that a Fubini word $\gamma$ is \emph{associated} with $v$ if either
\begin{itemize}
\item $v = 0^n$ and the only zero in $\gamma$ occurs at $\gamma_1$, or
\item $v \neq 0^n$ and $\gamma_i = 0$ if and only if $v_i = 1$. 
\end{itemize}
\end{defn}

\begin{defn}
A \emph{barred Fubini word} associated with $v$ is a Fubini word $\gamma$ associated with $v$ where we may place bars over certain entries. Specifically, the entry $\gamma_j$ may be barred if 
\begin{enumerate}
\item $\gamma_j > 0$, 
\item $\gamma_j$ is unique in $\gamma$, and
\item for each $i<j$ we have $\gamma_i < \gamma_j$, i.e.\ $\gamma_j$ is a left-to-right maximum in $\gamma$. 
\end{enumerate}
We denote the collection of barred Fubini words associated with $v$ by $\fubini_v$. 
\end{defn}

For example, 
\begin{align}
\fubini_0 &= \{0\} \\
\fubini_{00} &= \{01, 0\overline{1}\} \\
\fubini_{000} &= \{011, 012, 0\overline{1}2, 01\overline{2}, 0\overline{1}\overline{2}, 021, 0\overline{2}1 \} .
\end{align}
The sequence $|\fubini_{0^n}|$ for $n \in \mathbb{N}$ begins $1, 1, 2, 7, 35, 226,\ldots$ and seems to appear in the OEIS as A014307 \cite{oeis}. One way to define sequence A014307 is that it has exponential generating function
\begin{align}
\sqrt{\frac{e^z}{2 - e^z}} .
\end{align}
This sequence is given several combinatorial interpretations in \cite{ren}. It would be interesting to obtain a bijection between $\fubini_{0^n}$ and one of the collections of objects in \cite{ren}. See \cref{fig:fubini-words} for more examples of barred Fubini words.

\begin{figure}
\begin{center}
\begin{tabular}{c|c}
$v$	& $\fubini_v$ \\\hline
111	& $000$ \\
011	& $100, \overline{1}00$\\
101	& $010, 0\overline{1}0$\\
110	& $001, 00\overline{1}$\\
001 & $110, 120, 1\overline{2}0, \overline{1}20, \overline{1}\overline{2}0, 210, \overline{2}10 $\\
010	& $101, 102, 10\overline{2}, \overline{1}02, \overline{1}0\overline{2}, 201, \overline{2}01$\\
100	& $011, 012, 0\overline{1}2, 01\overline{2}, 0\overline{1}\overline{2}, 021, 0\overline{2}1  $ \\
000	& $011, 012, 0\overline{1}2, 01\overline{2}, 0\overline{1}\overline{2}, 021, 0\overline{2}1  $
\end{tabular}
\caption{We have listed the barred Fubini words $\fubini_v$ for each $v \in \{0,1\}^3$.}
\label{fig:fubini-words}
\end{center}
\end{figure}


Given a barred Fubini word $\gamma$ and a word $\pi \in \mathbb{Z}_{+}^n$, we use the same area statistic and modify the dinv statistic slightly:
\begin{align}
\area(\gamma) &= |\gamma| - \# \{1 \leq i \leq n : \gamma_i > 0\} \\
\dinv(\gamma, \pi) &= \# \{1 \leq i < j \leq n : \gamma_i = \gamma_j, \pi_i > \pi_j \} \\
&+ \#\{1 \leq i < j \leq n : \gamma_i + 1 = \gamma_j, \pi_i < \pi_j, \gamma_j \text{ is not barred} \} \nonumber.
\end{align}
We also let $\barstat(\gamma)$ be the number of barred entries in $\gamma$. We have the following result.

\begin{thm}
\label{thm:link-finite}
For $v \in \{0,1\}^n$, 
\begin{align}
\linkpoly_{v} &= \sum_{\substack{\gamma \in \fubini_v \\ \pi \in \mathbb{Z}_{+}^n}} q^{\area(\gamma) + \barstat(\gamma)} t^{\dinv(\gamma, \pi)} (1-q)^{-\barstat(\gamma)-\chi(v=0^n)} x^{\pi}
\end{align}
where $\chi$ of a statement is 1 if the statement is true and 0 if it is false.
\end{thm}

As in \cref{sec:infinite}, we give a formula for computing $f_v(q,a,t)$ directly. Given a barred Fubini word $\gamma$, we define
\begin{align}
\dinv_i(\gamma) &= \# \{j < i : \gamma_j = \gamma_i\} + \# \{j > i : \gamma_j  = \gamma_i + 1, \text{$\gamma_j$ is not barred} \}.
\end{align}

\begin{cor}
\label{cor:f-finite}
\begin{align}
f_v(q,a,t) &= \sum_{\gamma \in \fubini_v} q^{\area(\gamma) + \barstat(\gamma)} (1-q)^{-\barstat(\gamma) - \chi(v=0^n)} \prod_{i=1}^{n} \left(a + t^{\dinv_i(\gamma)} \right) 
\end{align}
\end{cor}

Let us use \cref{cor:f-finite} to compute $f_{00}(q,a,t)$ again. We note that $\fubini_{00} = \{01, 0\overline{1}\}$. Computing the weights for these vectors and then adding, we get
\begin{align}
f_{00}(q,a,t) &= \frac{(1+a)(a+t)}{1-q} + \frac{q(1+a)^2}{(1-q)^2} \\
&= \frac{q+t-qt + a(1+q+t-qt) + a^2}{(1-q)^2}
\end{align}
which is the same answer we computed at the end of \cref{sec:infinite}.

\section{Conjectures}
\label{sec:conjectures}

So far, we have used the inner product $\langle \linkpoly_v, e_{n-d} h_d\rangle$ to compute $f_v(q,a,t)$; one might wonder if there is any value in studying the full symmetric function $L_v$. In this section, we conjecture that the link symmetric function $L_v$ is closely related to the combinatorics of Macdonald polynomials, hinting at a stronger connection between Macdonald polynomials and link homology. Following \cite{elias-hogancamp}, we must first define a ``normalized'' version of the link symmetric function $\linkpoly_{v}$.

\begin{defn}
\begin{align}
\linknorm_{v} = \linknorm_v(x;q,t) = (1-q)^{n-|v|} \linkpoly_{v}(x; q, t). 
\end{align}
\end{defn}

We could also define $\linknorm_{v}$ in terms of diagrams; each box that contains a label from $\pi$ contributes an additional factor of $1-q$. \cref{thm:link-finite} implies that $\linknorm_{v}$ has coefficients in $\mathbb{Z}[q,t]$, whereas the coefficients of $\linkpoly_{v}$ are elements of $\mathbb{Z}[[q,t]]$. We conjecture that the normalized link symmetric function $\linknorm_{v}$ is closely connected to the Macdonald eigenoperators $\nabla$ and $\Delta$. 

The modified Macdonald polynomials $\widetilde{H}_{\mu}$ form a basis for the ring of symmetric functions with coefficients in $\mathbb{Q}(q,t)$. They can be defined via triangularity relations or combinatorially \cite{hhl, haglund-book}. Given a partition $\mu$, let $B_{\mu}$ be the alphabet of monomials $q^{i} t^j$ where $(i,j)$ ranges over the coordinates of the cells in the Ferrers diagram of $\mu$. We compute an example in \cref{fig:bmu}.

\begin{figure}
\begin{center}
\begin{displaymath}
\begin{ytableau}
t^2 \\
t	& qt	& q^2t  \\
1	&  q 	&  q^2	&  q^3
\end{ytableau}
\end{displaymath}
\caption{This is the Ferrers diagram of the partition $\mu = (4,3,1)$. In each cell we have written the monomial $q^i t^j$ that corresponds to the cell, yielding $B_{\mu} = \{1, q, q^2, q^3, t, qt, q^2 t, t^2 \}$.}
\label{fig:bmu}
\end{center}
\end{figure}

Given a symmetric function $F$ and a set of monomials $A = \{a_1, a_2, \ldots, a_n\}$, we let $F[A]$ be the result of setting $x_i = a_i$ for $1 \leq i \leq n$ and $x_i = 0$ for $i > n$. Then we define two operators on symmetric functions by setting, for $\mu \vdash n$,
\begin{align}
\Delta_F \widetilde{H}_{\mu} &= F \left[ B_{\mu} \right] \widetilde{H}_{\mu} \\
\nabla \widetilde{H}_{\mu} &= \Delta_{e_n} \widetilde{H}_{\mu}
\end{align}
and expanding linearly. Note that, for $\mu \vdash n$, $e_n[B_{\mu}]$ is simply the product of the $n$ monomials in $B_{\mu}$, which is sometime written $T_{\mu}$ in the literature.

\begin{conj}
\label{conj:p1^n}
\begin{align}
\label{p1^n1}
\nabla p_{1^n} &= \linknorm_{0^n} \\
\label{p1^n2}
\Delta_{e_{n-1}} p_{1^n} &= \sum_{\substack{ v \in \{0,1\}^n \\  |v| = 1}} \linknorm_{v}
\end{align}
In fact, both conjectures follow from the conjecture that
\begin{align}
\label{p1^n3}
\linknorm_{v0} &= \nabla p_1 \nabla^{-1} \linknorm_v.
\end{align}
\end{conj}

We should mention that Eugene Gorsky first noticed that the identity
\begin{align}
\sum_{a=0}^{d} \left\langle \nabla p_{1^n}, e_{n-d} h_d \right\rangle a^d &= (1-q)^n f_{0^n}(q,a,t)
\end{align}
seemed to hold and communicated this observation to the author via Jim Haglund. Gorsky's conjectured identity is a special case of \cref{conj:p1^n}. It is also interesting to note that the operator in \eqref{p1^n3} appears in the setting of the Rational Shuffle Conjecture as $-\mathbf{Q}_{1,1}$ \cite{rational-shuffle}.

As an example of our conjecture, we can use Sage to compute
\begin{align}
\left\langle \nabla p_{1,1}, p_{1,1} \right\rangle &= 1 + q + t - qt.
\end{align}
This expression should equal $\left\langle \linknorm_{00}, p_{1,1}\right\rangle$ by \cref{conj:p1^n}.  To compute this inner product using \cref{thm:link-finite}, we consider the barred Fubini words $01$ and $0\overline{1}$, each of which can receive labels $\pi = 12$ or $21$. The corresponding diagrams are

\vspace{10pt}

\begin{center}
\begin{tikzpicture}
\draw [step=0.5cm] (-0.001, 0) grid (0.5, 0);
\draw [step=0.5cm] (0.499, 0) grid (1, 0.5); 

\node at (0.25, -0.25) {1};
\node at (0.75, 0.25) {2};

\draw [step=0.5cm] (1.999, 0) grid (2.5, 0);
\draw [step=0.5cm] (2.499, 0) grid (3, 0.5); 

\node at (2.25, -0.25) {2};
\node at (2.75, 0.25) {1};

\draw [step=0.5cm] (3.999, 0) grid (4.5, 0);
\draw [step=0.5cm] (4.499, 0) grid (5, 0.5); 

\node at (4.25, -0.25) {1};
\node at (4.75, 0.25) {$\overline{2}$};

\draw [step=0.5cm] (5.999, 0) grid (6.5, 0);
\draw [step=0.5cm] (6.499, 0) grid (7, 0.5); 

\node at (6.25, -0.25) {2};
\node at (6.75, 0.25) {$\overline{1}$};
\end{tikzpicture}
\end{center}
where we have moved the bars from $\gamma_i$ to the corresponding $\pi_i$. The weights of these diagrams coming from \cref{thm:link-finite} are 
\begin{align}
\frac{t}{1-q} \qquad \quad \frac{1}{1-q} \qquad \quad  \frac{q}{(1-q)^2} \qquad \quad \frac{q}{(1-q)^2} 
\end{align}
respectively. After multiplying by the normalizing factor $(1-q)^2$ to go from $\linkpoly_{00}$ to $\linknorm_{00}$, we sum the resulting weights to get 
\begin{align}
(1-q)t + 1-q + q + q = 1 + q + t - qt
\end{align}
as desired.

After reading an earlier version of this paper, Fran\c{c}ois Bergeron contacted the author with the following additional conjectures.


\begin{conj}[Bergeron, 2016]
\label{conj:bergeron}
\begin{align}
\label{bergeron-append-0}
\linkpoly_{v0} &= \linkpoly_{1v} + q \linkpoly_{0v} \\
\label{bergeron2}
\linkpoly_{0^n} &= \sum_{v \in \{0,1\}^k} q^{n-|v|} \linkpoly_{v0^{n-k}} \\
t \left(\linkpoly_{u011v} - \linkpoly_{u101v} \right) &= \linkpoly_{u101v} - \linkpoly_{u110v} \\
\linknorm_{0^a 1^b 0^c} &= \nabla p_{1^c} \nabla^{-1} \widetilde{H}_{1^b} \nabla p_{1^a}  \\
\linkpoly_{1^a 0 1^b} &= \frac{t^a-1}{t^{a+b}-1} \left[ \nabla p_1 \nabla^{-1}, \widetilde{H}_{1^{a+b}} \right] + \widetilde{H}_{1^{a+b}} p_1 
\end{align}
where the  bracket represents the Lie bracket and operators are applied to 1 if nothing is explicitly specified. Bergeron also observed that $\linkpoly_v(x;q,1+t)$ is $e$-positive. (For more context on this last statement, see Section 4 of \cite{bergeron-open}.) 
\end{conj}

It is clear that \eqref{bergeron-append-0} implies \eqref{bergeron2}. We do not know of any other relations between these conjectures. We close with more open questions.
\begin{enumerate}
\item Is there a Macdonald eigenoperator expression for $\linknorm_{v}$ for other $v$?  Perhaps we can use ideas from the Rational Shuffle Conjecture \cite{rational-shuffle}, recently proved by Mellit \cite{mellit-rational}. In particular, Mellit's proof relies heavily on facts about toric braids, which seems relevant to the appearance of torus links in our setting.

\item Can we generalize our conjecture for $\nabla p_{1^n}$ to ``interpolate'' between our conjecture and the Shuffle Theorem \cite{carlsson-mellit}, or maybe the Square Paths Theorem \cite{leven-square}?
\end{enumerate}

\section{Acknowledgements}

The author would like to Ben Elias and Matt Hogancamp for their exciting paper and for use of \cref{fig:cv-braid}; Lyla Fadali for reading an earlier draft; Jim Haglund for editing and feedback; Eugene Gorsky for his comments and for the idea that Elias and Hogancamp's work could be related to Macdonald polynomials; and Fran\c{c}ois Bergeron for \cref{conj:bergeron} along with other helpful suggestions.

\printbibliography

\end{document}
